Author's response to reviews

Title: Building an ethical environment improves patient privacy and satisfaction in the crowded emergency department: A quasi-experimental study

Authors:

Yen-Ko Lin (yenko@ms16.hinet.net)
Wei-Che Lee (doctor.tezu@gmail.com)
Liang-Chi Kuo (lckaku@yahoo.com.tw)
Yuan-Chia Cheng (plkm111.cheng@msa.hinet.net)
Chia-Ju Lin (chiaju@kmu.edu.tw)
Hsing-Lin Lin (hsinglin2002@yahoo.com.tw)
Chao-Wen Chen (ytljwc@yahoo.com.tw)
Tsung-Ying Lin (cybersphinx@msn.com)

Version: 4 Date: 25 January 2013

Author's response to reviews: see over
Dear Editor Adrian Aldcroft, Dr. Marco De Ambrogi, and Reviewers,

We revised our manuscript entitled “Building an ethical environment improves patient privacy and satisfaction in the crowded emergency department: A quasi-experimental study” to *BMC Medical Ethics* for publication. The manuscript number is MS:7648023507873253.

Thanks for Editor Adrian Aldcroft, Dr. Marco De Ambrogi, and Reviewers’ precious time and reviews. We are appreciative of your valuable comments, and have responded to the questions and concerns you raised, revising our paper accordingly. The “Competing interests” has been added after the Conclusions, indicating that “The authors declare that they have no competing interests.” We have also checked the revised manuscript carefully and ensure the revised manuscript conforms to the journal style.

Sincerely,

Yen-Ko Lin, MD   January 25, 2013